\documentclass[preprint,draft]{elsarticle}
\usepackage{amsmath,amsthm}
\usepackage[english]{babel}
\usepackage{amsfonts}
\usepackage{latexsym}

\tolerance 300
\renewcommand{\thesubsection}{\arabic{subsection}}
\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{corollary}{Corollary}
\newcommand*{\cd}{(\cdot)}
\newcommand{\wx}{\widehat x}
\newcommand{\ov}{\overline}
\newcommand*{\la}{\langle}
\newcommand*{\ra}{\rangle}
\newcommand*{\LL}{\mathcal L}
\newcommand*{\wl}{\widehat\lambda}
\newcommand*{\lp}{L_p(\mathbb R)}
\newcommand*{\lt}{L_2(\mathbb R)}
\newcommand*{\iR}{\int_{\mathbb R}}
\newcommand*{\wdd}{\widetilde d}
\newcommand*{\vq}{\widetilde q}
\newcommand*{\wII}{\widetilde I}
\newcommand*{\wC}{\widetilde K}
\newcommand*{\ld}{L_2(\mathbb R^d)}
\newcommand*{\lpd}{L_p(\mathbb R^d)}
\newcommand*{\wL}{\widetilde L}
\newcommand*{\iRd}{\int_{\mathbb R^d}}
\newcommand*{\wM}{\widetilde M}

\DeclareMathOperator*{\sign}{sign}

\begin{document}
\begin{thebibliography}{11}

\bibitem{An} F.I.~Andrianov, Multidimensional analogues of Carlson's inequality and its generalizations, Izv. Vyssh. Uchebn. Zaved. Mat. 1 (56) (1960) 3--7 (in Russian).

\bibitem{Ar} V.V.~Arestov, Approximation of unbounded operators by bounded operators and related extremal problems, Russian Math. Surveys 51(6) (1996) 1093--1126.

\bibitem{B} S.~Barza, V.~Burenkov, J.~Pe\v cari\'c, L.-E.~Persson, Sharp multidimentional multiplicative inequalities for weighted $L_p$ spaces with homogeneous weights, Math. Ineq. Appl. 1 (1998) 53--67.

\bibitem{Ca} F.~Carlson, Une in\'egalit\'e, Ark. Mat. Astr. Fysik 25B (1934) 1--5.

\bibitem{Ha} J.~Hadamard, Sur le module maximum d'une fonction et ses d\'eriv\'ees, C. R. Soc. Math. France 41 (1914) 68--72.

\bibitem{HLP} G.G.~Hardy, J.E.~Littlewood, G.~P\'olya, Inequalities, Cambridge Univ. Press, Cambridge, 1934.

\bibitem{Ko} A.N.~Kolmogorov, On inequalities between the upper bounds of the successive derivatives of an arbitrary function on an infinite interval, Amer. Math Soc. Transl. 1 (4) (1949) 1-19.

\bibitem{La} E.~Landau, Ungleichungen f\"ur zweimal differenzierbar Funktionen, Proc. London. Math. Soc. 2 (13) (1913) 43--49.

\bibitem{Le} V.I.~Levin, Sharp constants in inequalities of Carlson type, Dokl. Akad. Nauk. SSSR 59 (1948) 635--638 (in Russian).

\bibitem{MO} G.G.~Magaril-Il'yaev, K.Yu.~Osipenko, Optimal recovery of functions and their derivatives from inaccurate information about the spectrum and inequalities for derivatives, Funct. Anal. Appl. 37 (3) (2003), 203--214.

\bibitem{MO1} G.G.~Magaril-Il'yaev, K.Yu.~Osipenko, Optimal recovery of values of functions and their derivatives from inaccurate data on the Fourier transform, Sb. Math. 195 (10) (2004) 1461--1476.
\bibitem{MS} G.G.~Magaril-Il'yaev, E.O.~Sivkova, Best recovery of the Laplace operator of a function from incomplete spectral data, Sb. Math. 203 (4) (2012) 569--580.

\bibitem{MM} A.A.~Melkman, C.A.~Micchelli, Optimal estimation of linear operators in Hilbert spaces from inaccurate data, SIAM J. Numer. Anal. 16 (1979) 87--105.

\bibitem{Os1} K.Yu.~Osipenko, Optimal recovery of linear operators in non-Euclidean metrics, Sb. Math. 205 (10) (2014) 1442--1472.

\bibitem{Os} K.Yu.~Osipenko, Optimal recovery of operators and multidimensional Carlson type inequalities, J. Complexity 32 (1) (2016) 53--73.

\bibitem{Si} E.O.~Sivkova, Best recovery of the Laplace operator of a function and sharp inequalities, J. Math. Sci. 209 (1) (2015) 130--137.

\bibitem{St} E.M.~Stein, Functions of exponential type, Ann. of Math. (2) 65 (3) (1957) 582--592.

\bibitem{Ta} L.V.~Taikov, Kolmogorov-type inequalities and the best formulas for numerical differentiation, Math. Notes 4 (2) (1968) 631--634.

\bibitem{TM} V.M.~Tikhomirov, G.G.~Magaril-Il'yaev, Inequalities for derivatives, Selected works of A.N.~Kolmogorov, Vol 1. Mathematics and Mechanics, Kluwer, Dordrecht, 442--446.




\end{thebibliography}

\begin{frontmatter}
\title{Inequalities for derivatives with the Fourier transform}
\author{K.~Yu.~Osipenko}%\fnref{myfootnote1}}
%\cortext[cor]{Corresponding author}
%\fntext[myfootnote1]{The research was carried out with the financial support of the Russian %Foundation for Basic Research (grant no.\ 17-01-00649)}
\address{Moscow State University,\\
Institute for Information Transmission Problems,
Russian Academy of Sciences, Moscow,\\
Moscow Aviation Institute (National Research University)}
\ead{kosipenko@yahoo.com}
\begin{abstract}
In this paper we study sharp constants in inequalities of the following form
$$\|x^{(k)}\cd\|_{L_q(\mathbb R)}\le K\|Fx\cd\|_{L_p(\mathbb R)}^\alpha\|x^{(n)}\cd\|_{L_r(\mathbb R)}^\beta,$$
where $Fx\cd$ is the Fourier transform of $x\cd$. The sharp value of $K$ in the general case (that is, for all $n\in\mathbb N$ and $0\le k<n$) was known only for $q=r=2$ and $p\ge2$. We obtain the sharp constant in the general case for $q=\infty$, $r=2$, and $1\le p\le\infty$. We also generalized this two cases on multidimensional situation. The sharp constants is obtained for fractional degrees of the Laplace operator $(-\Delta)^{k/2}$ and derivatives $D^\alpha$ of order $\alpha=(\alpha_1,\ldots,\alpha_d)\in\mathbb R^d_+$.
\end{abstract}

\begin{keyword}
inequality for derivatives \sep sharp constants \sep multidimensional inequalities for derivatives \sep Carlson type inequalities
\MSC[2010] 26A10 \sep 41A44 \sep 42A38
\end{keyword}

\end{frontmatter}

\section{Introduction}

Inequalities for derivatives of Kolmogorov-type on the real line are traditionally understood as multiplicative inequalities of the form
\begin{equation}\label{31}
\|x^{(k)}\cd\|_{L_q(\mathbb R)}\le K\|x\cd\|_{L_p(\mathbb R)}^\alpha\|x^{(n)}\cd\|_{L_r(\mathbb R)}^\beta,
\end{equation}
where $0\le k<n$ are integers, $1\le p,q,r\le\infty$, $\alpha,\beta\ge0$. It is assumed that the function $x\cd\in L_p(\mathbb R)$ has the $(n-1)$-st derivative which is locally absolutely continuous on $\mathbb R$ and $x^{(n)}\cd\in L_r(\mathbb R)$. The problem is to find the best possible constant $K$ (the smallest constant) in \eqref{31}. Inequalities \eqref{31} with the best possible constant we call the sharp inequalities.

The first sharp inequalities of type \eqref{31} were obtained by Landau \cite{La} (for the half-line $\mathbb R_+$, $n=2$, $k=1$, $p=q=r=\infty$) and Hadamard \cite{Ha} (for the line $\mathbb R$, $n=2$, $k=1$, $p=q=r=\infty$). In the late 30s Kolmogorov \cite{Ko} determined the exact constant in \eqref{31} for line, $p=q=r=\infty$ in the general case, i.e., for any $n\ge2$ and $0<k<n$. This result remains one of the most remarkable ones in this area, and the sharp inequalities of type \eqref{31} are often referred to as Kolmogorov's inequalities. Paying tribute to the first result in this area, these inequalities are also called Landau--Kolmogorov inequalities.

There are a lot of papers devoted to the search of the best constants in inequalities of such type (see survey papers \cite{TM} and \cite{Ar}). However, there are only three more complete results for the line comparable to Kolmogorov's were obtained ($p=q=r=2$ by Hardy, Littlewood, and P\'olya \cite{HLP}, $p=q=r=1$ by Stein \cite{St}, and $p=r=2$, $q=\infty$ by Taikov \cite{Ta}).

Sharp inequalities for derivatives are closely connected with the problems of optimal recovery of functions and their derivatives from an inaccurate information about function itself or its Fourier transform (see \cite{MM} and \cite{MO}). In the latter case sharp inequalities contain the Fourier transform of function instead of function. It turns out that inequalities of this type can be obtained even in a more general form. Namely, they are valid not only for all $n\in\mathbb N$ and $0\le k<n$ but for some family of metrics in which the Fourier transform of function is measured. One of the first result of such type was obtained in \cite{MO}.

Let $S$ be the Schwartz space of rapidly decreasing infinitely
differentiable functions on $\mathbb R$, $S'$ the dual space of
distributions, and $F\colon S'\to S'$ the Fourier transform. Let $1\le p\le
\infty$ and $n\in\mathbb N$. Set
$$X_p^n=\{\,x\in S':Fx\cd\in\lp,\ x^{(n)}\cd\in\lt\,\}.$$
It was proved in \cite{MO} that for all $x\cd\in X_p^n$, $2\le p\le\infty$, and all $0\le k<n$ the following sharp inequality holds:
\begin{equation}\label{hlp}
\|x^{(k)}\cd\|_{\lt}\le K(k,n,p)\|Fx\cd\|_{\lp}^{\frac{n-k}{n+1/2-1/p}}\|x^
{(n)}\cd\|_{\lt}^{\frac{k+1/2-1/p}{n+1/2-1/p}},
\end{equation}
where for $2<p\le\infty$
\begin{gather*}
K(k,n,p)=\sqrt{\frac{n+1/2-1/p}{k+1/2-1/p}}\left(\frac{\sqrt{k+1/2-1/p}B^{1
/2-1/p}}{\sqrt{2\pi}(n-k)^{1-1/p}}\right)^{\frac{n-k}{n+1/2-1/p}},\\
B=B\left(\frac{k+1/2-1/p}{(n-k)(1-2/p)},2\frac{1-1/p}{1-2/p}\right),
\end{gather*}
and $B(\cdot,\cdot)$ is the Eiler beta-function; for $p=2$
$$K(k,n,2)=\left(\frac1{2\pi}\right)^{\frac{n-k}{2n}}.$$
The case $p=2$ immediately follows from the Hardy--Littlewood--P\'olya inequality, so \eqref{hlp} may be considered as a generalized Hardy--Littlewood--P\'olya inequality.

This paper is devoted to the study of sharp inequalities of such type, including the multidimensional situation.

\section{The Carlson, Levin, and Taikov inequalities}

The Carlson inequality \cite{Ca}
$$\|x(t)\|_{L_1(\mathbb R_+)}\le\sqrt\pi\|x(t)\|_{L_2(\mathbb R_+)}^{1/2}
\|tx(t)\|_{L_2(\mathbb R_+)}^{1/2},\quad\mathbb R_+=[0,+\infty), $$
was generalized by many authors (see \cite{Le}, \cite{An}, \cite{B}, \cite{Os}).

In particular, it was proved by Levin \cite{Le} that for $p>1$, $q>1$, $\lambda>0$, $\mu>0$ the following sharp inequality holds:
\begin{equation}\label{Lev}
\|x(t)\|_{L_1(\mathbb R_+)}\le K\|t^{1-\frac{1+\lambda}p}x(t)\|_{L_p(\mathbb R_+)}^{pa}\|t^{1+\frac{\mu-1}q}x(t)\|_{L_q(\mathbb R_+)}^{qb},
\end{equation}
where
$$a=\frac\mu{p\mu+q\lambda},\quad b=\frac\lambda{p\mu+q\lambda},$$
and
$$K=\left(\frac1{pa}\right)^a\left(\frac1{qb}\right)^b
\left(\frac1{\lambda+\mu}B\left(\frac a{1-a-b},\frac b{1-a-b}\right)
\right)^{1-a-b}.$$

\begin{lemma}[\cite{B}]
Let $1<p,q<\infty$ and $\lambda,\mu>0$. The following sharp inequality holds:
\begin{equation}\label{lR}
\|x(t)\|_{L_1(\mathbb R)}\le2^{1-a-b}K\|t^{1-\frac{1+\lambda}p}x(t)\|_{\lp}^{pa}\|t^{1+\frac{\mu-1}q}x(t)\|_{L_q(\mathbb R)}^{qb}.
\end{equation}
\end{lemma}

\begin{proof}
For all positive $a_1,a_2,b_1,b_2$ we have (see \cite{HLP})
\begin{equation}\label{hl}
a_1^ab_1^b+a_2^ab_2^b\le2^{1-a-b}(a_1+a_2)^a(b_1+b_2)^b.
\end{equation}
Using this inequality we obtain
\begin{multline*}
\|x\cd\|_{L_1(\mathbb R)}=\int_{-\infty}^0|x(t)|\,dt+\int_0^\infty|x(t)|\,dt\\
\le K\biggl(\int_{-\infty}^0|t|^{p-1-\lambda}|x(t)|^p\,dt\biggr)^a
\biggl(\int_{-\infty}^0|t|^{q+\mu-1}|x(t)|^q\,dt\biggr)^b\\
+K\biggl(\int_0^\infty|t|^{p-1-\lambda}|x(t)|^p\,dt\biggr)^a
\biggl(\int_0^\infty|t|^{q+\mu-1}|x(t)|^q\,dt\biggr)^b\\
\le2^{1-a-b}K\|t^{1-\frac{1+\lambda}p}x(t)\|_{\lp}^{pa}\|t^{1+\frac{\mu-1}q}x(t)\|_{L_q(\mathbb R)}^{qb}.
\end{multline*}
The sharpness of this inequality follows from the sharpness of \eqref{Lev} and the fact that \eqref{hl} turns into equality for $a_1=a_2$ and $b_1=b_2$.
\end{proof}

\begin{theorem}
Let $n\in\mathbb N$, $0\le k<n$, $1\le p\le\infty$, and $p+k>1$. For all $x\cd\in X_p^n$ the following sharp inequality holds:
\begin{equation}\label{Ta}
\|x^{(k)}\cd\|_{L_\infty(\mathbb R)}\le K_1(k,n,p)\|Fx\cd\|_{\lp}^{\frac{n-k-1/2}{n+1/2-1/p}}
\|x^{(n)}\cd\|_{\lt}^{\frac{k+1-1/p}{n+1/2-1/p}},
\end{equation}
where
$$K_1=\left(\frac1\pi\right)^{1-b}\left(\frac1{pa}\right)^a\left(\frac1b\right)^b
\left(\frac1{n-k-1/2}B\left(\frac{1-b}{1-a-b},\frac b{1-a-b}\right)
\right)^{1-a-b}$$
and
$$a=\frac1p\cdot\frac{n-k-1/2}{n+1/2-1/p},\quad b=\frac12\cdot\frac{k+1-1/p}{n+1/2-1/p}.$$
\end{theorem}

\begin{proof}
Let $1<p<\infty$. Put $q=2$, $\lambda=p(1+k)-1$, $\mu=2n-2k-1$ in \eqref{lR} and change $x(t)$ by $y(\xi)$. Then we obtain
$$\|y(\xi)\|_{L_1(\mathbb R)}\le2^{1-a-b}K\|\xi^{-k}y(\xi)\|_{\lp}^{pa}\|\xi^{n-k}y(\xi)\|_{\lt}^{2b}.$$
If we take $y(\xi)=\xi^kFx(\xi)$ then we get
$$\iR|\xi|^k|Fx(\xi)|\,d\xi\le2^{1-a-b}K\biggl(\iR|Fx(\xi)|^p\,d\xi\biggr)^a\biggl(\iR|\xi|^{2n}|Fx(\xi)|^2
\,d\xi\biggr)^b.$$
By virtue of the Plancherel theorem we have
$$2\pi|x^{(k)}(t)|\le2^{1-a-b}K(2\pi)^b\|Fx\cd\|_{\lp}^{pa}\|x^{(n)}\cd\|_{\lt}^{2b}.$$
It follows from properties of beta-function that
$$B\left(\frac a{1-a-b},\frac b{1-a-b}\right)=\frac{a+b}aB\left(\frac{1-b}{1-a-b},\frac b{1-a-b}\right).$$
Substituting this value in the expression for $K$ we obtain \eqref{Ta}.

Suppose that $x\cd\in X_p^n$. If we take $\wx\cd$ such that
$$F\wx(\xi)=\varepsilon(\xi)e^{-it\xi}Fx(\xi),$$
where
$$\varepsilon(\xi)=\begin{cases}
(-i\sign\xi)^k\dfrac{\ov{Fx(\xi)}}{|Fx(\xi)|},&Fx(\xi)\ne0,\\
0,&Fx(\xi)=0,
\end{cases}$$
then
$$\biggl|\int_{\mathbb R}\psi(\xi)F\wx(\xi)e^{it\xi}\,d\xi\biggr|=\int_{\mathbb R}|\psi(\xi)Fx(\xi)|\,d\xi.$$

The cases $p=1,\infty$ were proved in \cite{MO1} (see Corollaries 1, 2).
\end{proof}

For $p=2$ inequality \eqref{Ta} has the form
$$\|x^{(k)}\cd\|_{L_\infty(\mathbb R)}\le K_2\left(\frac1{2\pi}\right)^{\frac{2n-2k-1}{4n}}\|Fx\cd\|_{\lt}^{\frac{2n-2k-1}{2n}}\|x^{(n)}
\cd\|_{\lt}^{\frac{2k+1}{2n}},$$
where
$$K_2=\left((2k+1)\sin\pi\dfrac{2k+1}{2n}\right)^{-1/2}\left(\frac{2k+1}
{2n-2k-1}\right)^{\frac{2n-2k-1}{4n}}.$$
In view of Plancherel's theorem we have
$$\|x^{(k)}\cd\|_{L_\infty(\mathbb R)}\le K_2\|x\cd\|_{\lt}^{\frac{2n-2k-1}{2n}}\|x^{(n)}
\cd\|_{\lt}^{\frac{2k+1}{2n}}.$$
This inequality was obtained by Taikov \cite{Ta} (without using of the Levin inequality).

\section{Multidimensional inequalities}

Let $d_1\cd$ and $d_2\cd$ be measurable functions on $\mathbb R^d$. Assume that $|d_1\cd|$ and $|d_2\cd|$ be homogenous functions of degrees $k$, $n$, respectively ($k$ and $n$ are not necessarily integer), $d_j(\xi)\ne0$, $j=1,2$, for almost all $\xi\in\mathbb R^d$. Consider the polar transformation
$$\arraycolsep=0.08em
\begin{array}{rcl}
\xi_1&=&\rho\cos\omega_1,\\
\xi_2&=&\rho\sin\omega_1\cos\omega_2,\\
\hdotsfor{3}\\
\xi_{d-1}&=&\rho\sin\omega_1\sin\omega_2\ldots\sin\omega_{d-2}\cos\omega_{d-1},\\
\xi_d&=&\rho\sin\omega_1\sin\omega_2\ldots\sin\omega_{d-2}\sin\omega_{d-1}.
\end{array}$$

Put $\omega=(\omega_1,\ldots,\omega_{d-1})$,
\begin{align*}
\wdd_1(\omega)&=|d_1(\cos\omega_1,\ldots,\sin\omega_1\sin\omega_2\ldots\sin\omega_{d-2}\sin
\omega_{d-1})|,\\
\wdd_2(\omega)&=|d_2(\cos\omega_1,\ldots,
\sin\omega_1\sin\omega_2\ldots\sin\omega_{d-2}\sin\omega_{d-1})|,\\
J(\omega)&=\sin^{d-2}\omega_1\sin^{d-3}\omega_2\ldots\sin\omega_{d-2}.
\end{align*}
Set
$$K_{pqr}(k,n,I)=\gamma^{-\frac\gamma p}(1-\gamma)^{-\frac{1-\gamma}r}\Biggl(
\frac{B\left(\vq\gamma/p+1,\vq(1-\gamma)/r\right)I}{r(n-k-d(1/q-1/r))}\Biggr)
^{1/\vq},$$
where
\begin{equation}\label{gq}
\gamma=\frac{n-k-d(1/q-1/r)}{n+d(1/r-1/p)},\quad\vq=\left(\frac1q-\frac{\gamma}p-
\frac{1-\gamma}r\right)^{-1}.
\end{equation}

We will use the following result which was proved in \cite[Theorem~2]{B} (see also \cite[Corollary~4]{Os}).

\begin{theorem}\label{M}
Let $1\le q<p,r<\infty$, $k>d(1/p-1/q)$, $n>k+d(1/q-1/r)$,
and
$$\wII_{qr}=\int_{\Pi_{d-1}}\frac{{\wdd_1\,}^{\vq}(\omega)}{{\wdd\,}_2^{\,\vq(1-\gamma)}
(\omega)}J(\omega)
\,d\omega<\infty,\quad\Pi_{d-1}=[0,\pi]^{d-2}\times[0,2\pi].$$
Then the following sharp inequality
\begin{equation}\label{CCC1}
\|d_1\cd x\cd\|_{L_q(\mathbb R^d)}\le K_{pqr}(k,n,\wII_{qr})\|x\cd\|_{L_p(\mathbb R^d)}^{\gamma}\|d_2\cd x\cd\|_{L_r(\mathbb R^d)}^{1-\gamma}
\end{equation}
holds.
\end{theorem}

For the case when $p=q$ we have

\begin{theorem}\label{TT1}
Let $1\le p=q<r<\infty$, $k>0$, $n>k+d(1/p-1/r)$, and
\begin{gather*}
I_{r1}=\int_{\mathbb R^d}|d_2(t)|^{\frac{pr}{p-r}}(|d_1(t)|-1)_+^{\frac p{r-p}}\,dt<\infty,\\
I_{r2}=\int_{\mathbb R^d}|d_2(t)|^{\frac{pr}{p-r}}(|d_1(t)|-1)_+^{\frac r{r-p}}\,dt<\infty
\end{gather*}
$(u_+=\max\{u,0\})$. Then the following sharp inequality
\begin{equation}\label{p1}
\|d_1\cd x\cd\|_{L_p(\mathbb R^d)}\le\wC(p,r)\|x\cd\|_{L_p(\mathbb R^d)}^{\frac{n-k-d(1/p-1/r)}{n-d(1/p-1/r)}}\|d_2\cd x\cd\|_{L_r(\mathbb R^d)}^{\frac k{n-d(1/p-1/r)}}
\end{equation}
holds, where
\begin{equation}\label{Kd}
\wC(p,r)=I_{r1}^{-\frac1p\frac{n-k-d(1/p-1/r)}{n-d(1/p-1/r)}}I_{r2}^{-\frac1r\frac k{n-d(1/p-1/r)}}(I_{r1}+I_{r2})^{1/p}.
\end{equation}
\end{theorem}

\begin{proof}
Consider the extremal problem
$$\|d_1\cd x\cd\|_{L_p(\mathbb R^d)}\to\max,\quad\|x\cd\|_{L_p(\mathbb R^d)}\le\delta,\quad
\|d_2\cd x\cd\|_{L_r(\mathbb R^d)}\le1.$$
Denote by $E$ the value of this problem. It follows from \cite[Theorem~2]{Os1} that
$$E=\left(\wl_1\delta^p+\frac rp\wl_2\right)^{1/p},$$
where $\wl_1>0$ satisfies the equation
\begin{multline*}
\biggl(\int_{\mathbb R^d}|d_2(\xi)|^{\frac{pr}{p-r}}(|d_1(\xi)|^p-\wl_1)_+^{\frac p{r-p}}\,d\xi\biggr)^{1/p}\\
=\delta\biggl(\int_{\mathbb R^d}|d_2(\xi)|^{\frac{pr}{p-r}}
(|d_1(\xi)|^p-\wl_1)_+^{\frac r{r-p}}\,d\xi\biggr)^{1/r}
\end{multline*}
and
$$\wl_2=\frac pr\delta^{p-r}\biggl(\int_{\mathbb R^d}|d_2(\xi)|^{\frac{pr}{p-r}}(|d_1(\xi)|^p-\wl_1)_+^{\frac p{r-p}}\,d\xi\biggr)^{\frac{r-p}p}.$$
Put $\wl_1=a^{pk}$ and change variables $\xi=at$. Then we obtain
$$a^{\frac{nr}{p-r}+\frac{kp}{r-p}+\frac dp}I_{r1}^{1/p}=\delta a^{\frac{np}{p-r}+\frac{kp}{r-p}+\frac dr}I_{r2}^{1/r}.$$
This implies that
$$a=\left(\frac{I_{r1}^{1/p}}{\delta I_{r2}^{1/r}}\right)^{\frac1{n-d(1/p-1/r)}}.$$
After the same change of variables we have
$$\wl_2=\frac pr\delta^{p-r}a^{-nr+kp+d(r/p-1)}I_{r1}^{r/p-1}.$$
Thus,
\begin{equation}\label{EE1}
E=\left(a^{pk}\delta^p+\delta^{p-r}a^{-nr+kp+d(r/p-1)}I_{r1}^{r/p-1}\right)^{1/p}=
\delta^{\frac{n-k-d(1/p-1/r)}{n-d(1/p-1/r)}}\wC(p,r).
\end{equation}

Let $\|x\cd\|_{L_p(\mathbb R^d)}<\infty$, $\|d_2\cd x\cd\|_{L_r(\mathbb R^d)}<\infty$, and $\|d_2\cd x\cd\|_{L_r(\mathbb R^d)}\ne0$. Put
$$\wx(t)=\frac{x(t)}{\|d_2\cd x\cd\|_{L_r(\mathbb R^d)}},\quad\delta=\|\wx\cd\|_{L_p(\mathbb R^d)}.$$
Then in view of \eqref{EE1} we have the sharp inequality
$$\|d_1\cd\wx\cd\|_{L_p(\mathbb R^d)}\le\wC(p,r)\|\wx\cd\|_{L_p(\mathbb R^d)}^{\frac{n-k-d(1/p-1/r)}{n-d(1/p-1/r)}}.$$
Passing to $x\cd$ we obtain \eqref{p1}.
\end{proof}

Set
$$Y_p=\{\,x\cd\in\ld:Fx\cd\in\lpd,\ d_2\cd Fx\cd\in\ld\,\}.$$
We define the operator $D_2\colon Y_p\to\ld$ as follows
\begin{equation}\label{D2}
D_2x\cd=F^{-1}(d_2\cd Fx\cd)\cd.
\end{equation}
Assume that $d_1\cd Fx\cd\in\ld$ for all $x\cd\in Y_p$. Put
\begin{equation}\label{D1}
D_1x\cd=F^{-1}(d_1\cd Fx\cd)\cd.
\end{equation}
We also assume that $D_1x\cd\in C(\mathbb R^d)$ for all $x\cd\in Y_p$.

Set
\begin{equation}\label{mpq}
M_{pq}(k,n,I)=\frac{K_{pq2}(k,n,I)}{(2\pi)^{d\frac{2n/q-k+(1-2/q)d/p}{2n+d(1-2/p)}}}.
\end{equation}

\begin{theorem}\label{TT}
Let $1\le p<\infty$, $k>d(1/p-1)$, and $n>k+d/2$. If $p>1$ and $\wII_{12}<\infty$ then for all
$x\cd\in Y_p$ the sharp inequality
\begin{equation}\label{D12}
\|D_1x\cd\|_{L_\infty(\mathbb R^d)}\le M_{p1}(k,n,\wII_{12})\|Fx\cd\|_{\lpd}^
{\frac{n-k-d/2}{n+d(1/2-1/p)}}\|D_2x\cd\|_{\ld}^{\frac{k+d(1-1/p)}{n+d(1/2-1/p)}}
\end{equation}
holds. If $p=1$ and $I_{21},I_{22}<\infty$ then for all $x\cd\in Y_1$ the sharp inequality
\begin{equation}\label{D122}
\|D_1x\cd\|_{L_\infty(\mathbb R^d)}\le\frac{\wC(1,2)}{(2\pi)^{d\frac{2n-k-d}{2n-d}}}\|Fx\cd\|_{L_1(\mathbb R^d)}^{\frac{2n-2k-d}{2n-d}}
\|D_2x\cd\|_{\ld}^{\frac{2k}{2n-d}}
\end{equation}
holds.
\end{theorem}

\begin{proof}
We have
$$D_jx(t)=\frac1{(2\pi)^d}\int_{\mathbb R^d}d_j(\xi)Fx(\xi)e^{i\la t,\xi\ra}\,d\xi,\quad j=1,2,$$
where $t=(t_1,\ldots,t_d)$, $\xi=(\xi_1,\ldots\xi_d)$, and $\la t,\xi\ra=t_1\xi_1+\ldots+t_d\xi_d$.
It follows from Theorem~\ref{M} that for $q=1$ and $r=2$ the following sharp inequality
$$\|d_1\cd Fx\cd\|_{L_1(\mathbb R^d)}\le K_{p12}(k,n,\wII_{12})\|Fx\cd\|_{\lpd}^\gamma
\|d_2\cd Fx\cd\|_{\ld}^{1-\gamma}$$
holds. Taking into account the Plancherel theorem we obtain \eqref{D12}. Using similar arguments from Theorem~\ref{TT1} we obtain \eqref{D122}.

For any $x\cd\in Y_p$ we take $\wx\cd$ such that
$$F\wx(\xi)=\varepsilon(\xi)e^{-i\la t,\xi\ra}Fx(\xi),$$
where
$$\varepsilon(\xi)=\begin{cases}
\dfrac{\ov{d_1(\xi)}\ov{Fx(\xi)}}{|d_1(\xi)Fx(\xi)|},&d_1(\xi)Fx(\xi)\ne0,\\
0,&d_1(\xi)Fx(\xi)=0.
\end{cases}$$
Then
$$\biggl|\int_{\mathbb R^d}d_1(\xi)F\wx(\xi)e^{i\la t,\xi\ra}\,d\xi\biggr|=\int_{\mathbb R^d}|d_1(\xi)Fx(\xi)|\,d\xi.$$
This implies that inequalities \eqref{D12} and \eqref{D122} are sharp.
\end{proof}

\section{Multidimensional generalizations of the Taikov inequality}

Consider some examples. Define the operator $(-\Delta)^{n/2}$, $n\ge0$, as follows
$$(-\Delta)^{n/2}x\cd=F^{-1}(|\xi|^n Fx(\xi))\cd,\quad|\xi|=\sqrt{\xi_1^2+\ldots+\xi_d^2}.$$
Put $d_1(\xi)=|\xi|^k$ and $d_2(\xi)=|\xi|^n$. Then operators $D_1$ and $D_2$ defined above coincide with $(-\Delta)^{k/2}$ and $(-\Delta)^{n/2}$.

\begin{corollary}
Let $1\le p\le\infty$, $k>d(1/p-1)$, and $n>k+d/2$. Then the sharp inequality
\begin{multline}\label{Ta11}
\|(-\Delta)^{k/2}x\cd\|_{L_\infty(\mathbb R^d)}\le M_{p1}(k,n,I_0)\\
\times\|Fx\cd\|_{\lpd}^{\frac{n-k-d/2}{n+d(1/2-1/p)}}
\|(-\Delta)^{n/2}x\cd\|_{\ld}^{\frac{k+d(1-1/p)}{n+d(1/2-1/p)}}
\end{multline}
holds, where
$$I_0=\frac{2\pi^{d/2}}{\Gamma(d/2)}.$$
\end{corollary}

\begin{proof}
Let $1<p<\infty$. Since $\wdd_1(\omega)=\wdd_2(\omega)=1$ we have
\begin{equation}\label{ii}
\wII_{12}=\int_{\Pi_{d-1}}J(\omega)\,d\omega=\frac{2\pi^{d/2}}{\Gamma(d/2)}.
\end{equation}
Now \eqref{Ta11} immediately follows from Theorem~\ref{TT}.

Let $p=1$. We have
$$I_{21}=\int_{|t|>1}|t|^{-2n}(|t|^k-1)\,dt,\quad
I_{22}=\int_{|t|>1}|t|^{-2n}(|t|^k-1)^2\,dt.$$
Passing to the polar transformation we get
$$I_{21}=\int_1^{+\infty}\rho^{-2n}(\rho^k-1)\rho^{d-1}\,d\rho\wII_{12}
=\frac{k\wII_{12}}{(2n-d)(2n-k-d)},$$
where $\wII_{12}$ is defined by \eqref{ii}. The analogous calculations give
$$I_{22}=\frac{2k^2\wII_{12}}{(2n-d)(2n-k-d)(2n-2k-d)}.$$
Substituting these values in \eqref{Kd} we obtain that
$$\wC(1,2)=\left(\frac{2n-d}{2n-2k-d}\right)^{\frac{2n-k-d}{2n-d}}
\left(\frac{\pi^{d/2}}{\Gamma(d/2)(2n-k-d)}\right)^{\frac k{2n-d}}.$$
It is easy to verify that
$$\wC(1,2)=K_{112}(k,n,I_0).$$

The case $p=\infty$ was obtained in \cite{Si}.
\end{proof}

Note that the value of $M_{21}(k,n,I_0)$ was calculated in \cite{Os}. Constants $M_{21}(k,n,I_0)$ and $M_{\infty1}(k,n,I_0)$ (as $M_{11}(k,n,I_0)$) can be written in a simpler form
\begin{gather*}
M_{21}(k,n,I_0)=\left(\frac{2k+d}{2n-2k-d}\right)^{\frac{2n-2k-d}{4n}}\left(\frac{\gamma_d}\pi(2k+d)
\sin\frac{2k+d}{2n}\pi\right)^{-1/2},\\
M_{\infty1}(k,n,I_0)=\frac{(n+d/2)^{\frac{k+d}{2n+d}}}{k+d}\left(\frac{2n-k}
{\gamma_d(2n-2k-d)}\right)^{\frac{2n-k}{2n+d}},
\end{gather*}
where
$$\gamma_d=2^{d-1}\pi^{d/2-1}\Gamma(d/2).$$

Consider one more example. Let $\alpha=(\alpha_1,\ldots,\alpha_d)\in\mathbb R_+^d$. We define $D^\alpha$ (the derivative of order $\alpha$) as follows
$$D^\alpha x\cd=F^{-1}((i\xi)^\alpha Fx(\xi))\cd,$$
where $(i\xi)^\alpha=(i\xi_1)^{\alpha_1}\ldots(i\xi_d)^{\alpha_d}$. Let $D_1=D^\alpha$ and $D_2=(-\Delta)^{n/2}$ (thus, $d_1(\xi)=(i\xi)^\alpha$ and $d_2(\xi)=|\xi|^n$).

Set $|\alpha|=\alpha_1+\ldots+\alpha_d$,
$$\vq_1=\frac{n+d(1/2-1/p)}{n(1-1/p)-|\alpha|(1/2-1/p)}.$$

\begin{theorem}
Let $1\le p\le\infty$, $|\alpha|\ge0$ if $p>1$ and $|\alpha|>0$ if $p=1$, $n>|\alpha|+d/2$. Then the sharp inequality
\begin{multline}\label{DA}
\|D^\alpha x\cd\|_{L_\infty(\mathbb R^d)}\le M_{p1}(|\alpha|,n,I_\alpha)\\
\times\|Fx\cd\|_{\lpd}^{\frac{n-|\alpha|-d/2}{n+d(1/2-1/p)}}
\|(-\Delta)^{n/2}x\cd\|_{\ld}^{\frac{|\alpha|+d(1-1/p)}{n+d(1/2-1/p)}}
\end{multline}
holds, where
$$I_\alpha=2\frac{\Gamma((\alpha_1\vq_1+1)/2)
\ldots\Gamma((\alpha_d\vq_1+1)/2)}{\Gamma((|\alpha|\vq_1+d)/2)}.$$
\end{theorem}

\begin{proof}
Let $1<p<\infty$. It follows from Theorem~\ref{TT} that sharp inequality \eqref{DA} holds with the constant $M_{p1}(|\alpha|,n,\wII)$, where
$$\wII=\int_{\Pi_{d-1}}|\cos\omega_1|^{\alpha_1\vq_1}\ldots|
\sin\omega_1\sin\omega_2\ldots\sin\omega_{d-2}\sin\omega_{d-1}|^{\alpha_d\vq_1}J(\omega)\,d\omega.$$
From the well-known Dirichlet formula we have
$$\int_{\substack{\xi_1\ge0,\ldots,\xi_d\ge0\\
\xi_1^2+\ldots+\xi_d^2\le1}}\xi_1^{p_1-1}\ldots\xi_d^{p_d-1}\,d\xi_1\ldots d\xi_d
=\frac{\Gamma(p_1/2)\ldots\Gamma(p_d/2)}
{2^d\Gamma(p_1/2+\ldots+p_d/2+1)},$$
$p_1,\ldots,p_d>0$. Passing to the polar transformation we obtain
$$\int_{\Pi_{d-1}}\Phi(\omega,p_1,\ldots,p_d)J(\omega)\,d\omega=2\frac{\Gamma(p_1/2)
\ldots\Gamma(p_d/2)}{\Gamma(p_1/2+\ldots+p_d/2)},$$
where
$$\Phi(\omega,p_1,\ldots,p_d)=|\cos\omega_1|^{p_1-1}\ldots
|\sin\omega_1\sin\omega_2\cdots\sin\omega_{d-2}\sin\omega_{d-1}|^{p_d-1}.$$
Consequently, $\wII=I_\alpha$.

Now let $p=1$. Let us use Theorem~\ref{TT1}. We have
\begin{gather*}
I_{21}=\int_{|t^\alpha|>1}|t|^{-2n}(|t^\alpha|-1)\,dt=J_{\alpha1}-J_{\alpha0},\\
I_{22}=\int_{|t^\alpha|>1}|t|^{-2n}(|t^\alpha|-1)^2\,dt=J_{\alpha2}-2J_{\alpha1}+J_{\alpha0},
\end{gather*}
where
$$J_{\alpha c}=\int_{|t^\alpha|>1}|t|^{-2n}|t^{c\alpha}|\,dt.$$
Passing to the polar transformation we have
\begin{multline*}
J_{\alpha c}=\int_{\Pi_{d-1}}\Phi(\omega,c\alpha_1+1,\ldots,c\alpha_d+1)
J(\omega)\,d\omega\int_{\Phi_0(\omega)}^{+\infty}
\rho^{-2n+c|\alpha|+d-1}\,d\rho\\
=\frac1{2n-c|\alpha|-d}\int_{\Pi_{d-1}}\Phi(\omega,\beta_1,\ldots,\beta_d)J(\omega)\,d\omega
=\frac{I_n(\alpha,d)}{2n-c|\alpha|-d},
\end{multline*}
where $\Phi_0(\omega)=\Phi(\omega,-\alpha_1/|\alpha|+1,\ldots,-\alpha_d/|\alpha|+1)$, $\beta_j=(2n-d)\alpha_j/|\alpha|+1$, $j=1,\ldots,d$, and
$$I_n(\alpha,d)=\frac2{\Gamma(n)}\Gamma\left(\alpha_1\frac{n-d/2}{|\alpha|}+1/2\right)\ldots
\Gamma\left(\alpha_d\frac{n-d/2}{|\alpha|}+1/2\right).$$
Thus,
\begin{gather*}
I_{21}=\frac{|\alpha|}{(2n-d)(2n-|\alpha|-d)}I_n(\alpha,d),\\
I_{22}=\frac{2|\alpha|^2}{(2n-d)(2n-|\alpha|-d)(2n-2|\alpha|-d)}I_n(\alpha,d).
\end{gather*}
Substituting these values in \eqref{Kd} we obtain
$$\wC(1,2)=\left(\frac{2n-d}
{2n-2|\alpha|-d}\right)^{\frac{2n-|\alpha|-d}{2n-d}}
\left(\frac{2I_n(\alpha,d)}{2n-|\alpha|-d}\right)^{\frac{|\alpha|}{2n-d}}.$$
It may be easily verified that $\wC(1,2)=K_{122}(|\alpha|,n,I_\alpha)$.

Now let $p=\infty$. Consider the extremal problem
\begin{equation}\label{extD}
|D^\alpha x(0)|\to\max,\quad\|Fx\cd\|_{L_\infty(\mathbb R^d)}\le\delta,\quad\|(-\Delta)^{n/2}x\cd\|_{\ld}\le1.
\end{equation}
Set
$$s_\alpha(\xi)=\begin{cases}\dfrac{(i\xi)^\alpha}{|\xi^\alpha|},&\xi^\alpha\ne0,\\
1,&\xi^\alpha=0\end{cases}$$
(we recall that $\xi^\alpha=\xi_1^{\alpha_1}\ldots\xi_d^{\alpha_d}$ and $|\xi^\alpha|=|\xi_1|^{\alpha_1}\ldots|\xi_d|^{\alpha_d}$). Let $\wx\cd$ be such that
$$F\wx(\xi)=\begin{cases}\delta\ov{s_\alpha(\xi)},&|\xi^\alpha|\ge\lambda|\xi|^{2n},\\
\dfrac\delta\lambda\dfrac{\ov{(i\xi)^\alpha}}{|\xi|^{2n}},&|\xi^\alpha|<\lambda|\xi|^{2n},
\end{cases}$$
where $\lambda>0$. We prove that for all $x\cd\in Y_\infty$ (for $d_2(\xi)=|\xi|^n$) the following equality
\begin{multline}\label{Toz}
D^\alpha x(0)=\frac1{(2\pi)^d}\int_{|\xi^\alpha|\ge\lambda|\xi|^{2n}}((i\xi)^\alpha-\lambda s_\alpha(\xi)|\xi|^{2n})Fx(\xi)\,d\xi\\
+\frac\lambda\delta\int_{\mathbb R^d}(-\Delta)^{n/2}x(t)\ov{(-\Delta)^{n/2}\wx(t)}\,dt
\end{multline}
holds. Indeed, in view of Plancherel's theorem we have
$$\int_{\mathbb R^d}(-\Delta)^{n/2}x(t)\ov{(-\Delta)^{n/2}\wx(t)}\,dt=\frac1{(2\pi)^d}\int_{\mathbb R^d}|\xi|^{2n}Fx(\xi)\ov{F\wx(\xi)}\,d\xi.$$
Consequently,
\begin{multline*}
\frac1{(2\pi)^d}\int_{|\xi^\alpha|\ge\lambda|\xi|^{2n}}((i\xi)^\alpha-\lambda s_\alpha(\xi)|\xi|^{2n})Fx(\xi)\,d\xi\\
+\frac\lambda\delta\frac1{(2\pi)^d}\int_{\mathbb R^d}|\xi|^{2n}Fx(\xi)\ov{F\wx(\xi)}\,d\xi\\
=\frac1{(2\pi)^d}\int_{|\xi^\alpha|\ge\lambda|\xi|^{2n}}((i\xi)^\alpha-\lambda s_\alpha(\xi)|\xi|^{2n})Fx(\xi)\,d\xi\\+\frac1{(2\pi)^d}\int_{|\xi^\alpha|\ge\lambda|\xi|^{2n}}
\lambda s_\alpha(\xi)|\xi|^{2n}Fx(\xi)\,d\xi+\frac1{(2\pi)^d}\int_{|\xi^\alpha|<\lambda|\xi|^{2n}}
(i\xi)^\alpha Fx(\xi)\,d\xi\\
=\frac1{(2\pi)^d}\int_{\mathbb R^d}(i\xi)^\alpha Fx(\xi)\,d\xi=D^\alpha x(0).
\end{multline*}
Choose $\lambda$ from the condition $\|(-\Delta)^{n/2}\wx\cd\|_{\ld}=1$. Then we get the following equation to find $\lambda$
$$\frac{\delta^2}{(2\pi)^d}\int_{|\xi^\alpha|\ge\lambda|\xi|^{2n}}|\xi|^{2n}\,d\xi+\frac{\delta^2}
{(2\pi)^d\lambda^2}\int_{|\xi^\alpha|<\lambda|\xi|^{2n}}\frac{|\xi^{2\alpha}|}
{|\xi|^{2n}}\,d\xi=1.$$
After passing to the polar transformation we obtain
\begin{multline*}
\frac{\delta^2}{(2\pi)^d}\int_{\Pi_{d-1}}J(\omega)\,d\omega\int_0^{\Phi_1(\omega)}
\rho^{2n+d-1}\,d\rho\\
+\frac{\delta^2\lambda^{-2}}{(2\pi)^d}\int_{\Pi_{d-1}}\Phi(\omega,2\alpha_1+1,\ldots,2\alpha_d+1)
J(\omega)\,d\omega\int_{\Phi_1(\omega)}^{+\infty}\rho^{-2n+2|\alpha|+d-1}\,d\rho\\
=1,
\end{multline*}
where
$$\Phi_1(\omega)=\lambda^{-\frac1{2n-|\alpha|}}\Phi\left(\omega,\frac{\alpha_1}{2n-|\alpha|}+1,
\ldots,\frac{\alpha_d}{2n-|\alpha|}+1\right).$$
Further, we arrive at the equation
$$\frac{\delta^2}{(2\pi)^d}\lambda^{-\frac{2n+d}{2n-|\alpha|}}
\frac{4n-2|\alpha|}{(2n+d)(2n-2|\alpha|-d)}J_0=1,$$
in which
\begin{multline*}
J_0=\int_{\Pi_{d-1}}\Phi\left(\omega,\alpha_1\frac{2n+d}{2n-|\alpha|}+1,\ldots,
\alpha_d\frac{2n+d}{2n-|\alpha|}+1\right)J(\omega)\,d\omega\\
=2\frac{\Gamma\left(\alpha_1\dfrac{2n+d}{4n-2|\alpha|}+1/2\right)\ldots
\Gamma\left(\alpha_d\dfrac{2n+d}{4n-2|\alpha|}+1/2\right)}{\Gamma\left(n\dfrac{|\alpha|+d}{2n-|\alpha|}
\right)}.
\end{multline*}
Thus,
\begin{equation}\label{lam}
\lambda=\left(\frac{\delta^2(4n-2|\alpha|)}{(2\pi)^d(2n+d)(2n-2|\alpha|-d)}J_0
\right)^{\frac{2n-|\alpha|}{2n+d}}.
\end{equation}

In view of \eqref{Toz} for any admissible function for problem \eqref{extD} we have
\begin{equation}\label{Dal}
|D^\alpha x(0)|\le\frac\delta{(2\pi)^d}\int_{|\xi^\alpha|\ge\lambda|\xi|^{2n}}(|\xi^\alpha|-\lambda|\xi|^{2n})
\,d\xi+\frac\lambda\delta.
\end{equation}
Since for $\wx\cd$ the inequality terns to equality inequality \eqref{Dal} is sharp.
We have
\begin{multline*}
\int_{|\xi^\alpha|\ge\lambda|\xi|^{2n}}(|\xi^\alpha|-\lambda|\xi|^{2n})
\,d\xi\\
=\int_{\Pi_{d-1}}\Phi(\omega,\alpha_1+1,\ldots,\alpha_d+1)J(\omega)\,d\omega
\int_0^{\Phi_1(\omega)}
\rho^{|\alpha|+d-1}\,d\rho\\
-\lambda\int_{\Pi_{d-1}}J(\omega)
\,d\omega\int_0^{\Phi_1(\omega)}
\rho^{2n+d-1}\,d\rho
=\lambda^{-\frac{|\alpha|+d}{2n-|\alpha|}}\frac{2n-|\alpha|}{(2n+d)(|\alpha|+d)}J_0.
\end{multline*}
Substituting this value in \eqref{Dal}, taking into account \eqref{lam}, we obtain the sharp inequality
$$|D^\alpha x(0)|\le K\delta^{\frac{n-|\alpha|-d/2}{n+d/2}},$$
where
$$K=\frac{n+d/2}{|\alpha|+d/2}\left(\frac{(n-|\alpha|/2)J_0}{(2\pi)^d(n+d/2)(n-|\alpha|-d/2)}
\right)^{\frac{n-|\alpha|/2}{n+d/2}}.$$
This implies the sharp inequality
$$|D^\alpha x(0)|\le K\|Fx\cd\|_{L_\infty(\mathbb R^d)}^{\frac{n-|\alpha|-d/2}{n+d/2}}
\|(-\Delta)^{n/2}x\cd\|_{\ld}^{\frac{|\alpha|+d}{n+d/2}}.$$
Due to the invariance with respect to the shift we get
$$\|D^\alpha x\cd\|_{L_\infty(\mathbb R^d)}\le K\|Fx\cd\|_{L_\infty(\mathbb R^d)}^{\frac{n-|\alpha|-d/2}{n+d/2}}
\|(-\Delta)^{n/2}x\cd\|_{\ld}^{\frac{|\alpha|+d}{n+d/2}}.$$
It is easy to check that $K=M_{\infty1}(|\alpha|,n,I_\alpha)$.
\end{proof}

For $p=2$ the constant $M_{21}(|\alpha|,n,I_\alpha)$ was found in \cite{Os}. It can be written in the form
$$M_{21}(|\alpha|,n,I_\alpha)=\left(\frac{2|\alpha|+d}{2n-2|\alpha|-d}
\right)^{\frac{2n-2|\alpha|-d}{4n}}
\left(\frac{2|\alpha|+d}{\Gamma_{\alpha d}}
\sin\frac{2|\alpha|+d}{2n}\pi\right)^{-1/2},$$
where
$$\Gamma_{\alpha d}=\frac{\Gamma(\alpha_1+1/2)\ldots\Gamma(\alpha_d+1/2)}
{(2\pi)^{d-1}\Gamma(|\alpha|+d/2)}.$$

\section{Multidimensional generalizations of the Hardy--Littlewood--P\'olya inequality}

Assume that functions $d_1\cd$ and $d_2\cd$ satisfy the conditions stated at the beginning of paragraph~3. We obtain an analog of Theorem~\ref{TT1} for the case when $q=r$.

\begin{theorem}\label{TTq}
Let $1\le q=r<p<\infty$, $k\ge0$, $n>k$, and
\begin{gather*}
S_{q1}=\int_{\mathbb R^d}(|d_1(t)|^q-|d_2(t)|^q)_+^{\frac p{p-q}}\,dt<\infty,\\
S_{q2}=\int_{\mathbb R^d}|d_2(t)|^q(|d_1(t)|^q-|d_2(t)|^q)_+^{\frac q{p-q}}\,dt<\infty.
\end{gather*}
Then the sharp inequality
\begin{equation}\label{p11}
\|d_1\cd x\cd\|_{L_q(\mathbb R^d)}\le L_{pq}\|x\cd\|_{L_p(\mathbb R^d)}^{\frac{n-k}{n+d(1/q-1/p)}}\|d_2\cd x\cd\|_{L_q(\mathbb R^d)}^{\frac{k+d(1/q-1/p)}{n+d(1/q-1/p)}}
\end{equation}
holds, where
$$L_{pq}=S_{q1}^{-\frac1p\frac{n-k}{n+d(1/q-1/p)}}S_{q2}^{-\frac1q\frac{k+d(1/q-1/p)}
{n+d(1/q-1/p)}}(S_{q1}+S_{q2})^{1/q}.$$
\end{theorem}

\begin{proof}
Consider the extremal problem
$$\|d_1\cd x\cd\|_{L_q(\mathbb R^d)}\to\max,\quad\|x\cd\|_{L_p(\mathbb R^d)}\le\delta,\quad
\|d_2\cd x\cd\|_{L_q(\mathbb R^d)}\le1.$$
Denote by $E$ the value of this problem. It follows from \cite[Theorem~1]{Os1} that
$$E=\left(\frac pq\wl_1\delta^p+\wl_2\right)^{1/q},$$
where $\wl_2>0$ satisfies the equation
\begin{multline*}
\biggl(\int_{\mathbb R^d}\left(|d_1(\xi)|^q-\wl_2|d_2(\xi)|^q\right)_+^{\frac p{p-q}}\,d\xi\biggr)^{1/p}\\
=\delta\biggl(\int_{\mathbb R^d}|d_2(\xi)|^q\left(|d_1(\xi)|^q-\wl_2|d_2(\xi)|^q\right)_+^{\frac q{p-q}}\,d\xi\biggr)^{1/q}
\end{multline*}
and
$$\wl_1=\frac qp\delta^{q-p}\biggl(\int_{\mathbb R^d}\left(|d_1(\xi)|^q-\wl_2|d_2(\xi)|^q\right)_+^{\frac p{p-q}}\,d\xi\biggr)^{\frac{p-q}p}.$$
Put $\wl_2=a^{(k-n)q}$ and change variables $\xi=at$. Then we obtain
$$a^{\frac{kq}{p-q}+\frac dp}S_{q1}^{1/p}=\delta a^{n+\frac{kq}{p-q}+\frac dq}S_{q2}^{1/q}.$$
This implies that
$$a=\left(\frac{S_{q1}^{1/p}}{\delta S_{q2}^{1/q}}\right)^{\frac1{n+d(1/q-1/p)}}.$$
After the same change of variables we have
$$\wl_1=\frac qp\delta^{q-p}a^{q(k+d(1/q-1/p))}S_{q1}^{\frac{p-q}p}.$$
Thus,
$$E=L_{pq}\delta^{\frac{n-k}{n+d(1/q-1/p)}}.$$
By arguments analogous to those that were used in the proof of Theorem~\ref{TT1} we obtain sharp inequality \eqref{p11}.
\end{proof}

Consider the operators $D_1$ and $D_2$ defined by \eqref{D1} and \eqref{D2}. It follows from \eqref{p11} that for all $x\cd\in Y_p$ we have
$$\|d_1\cd Fx\cd\|_{\ld}\le L_{p2}\|Fx\cd\|_{L_p(\mathbb R^d)}^{\frac{n-k}{n+d(1/2-1/p)}}\|d_2\cd Fx\cd\|_{\ld}^{\frac{k+d(1/2-1/p)}{n+d(1/2-1/p)}}.$$
In view of Plancherel's theorem we obtain
\begin{equation}\label{DH}
\|D_1x\cd\|_{\ld}\le\wL_{dp}\|Fx\cd\|_{L_p(\mathbb R^d)}^{\frac{n-k}{n+d(1/2-1/p)}}\|D_2x\cd\|_{\ld}^{\frac{k+d(1/2-1/p)}{n+d(1/2-1/p)}},
\end{equation}
where
$$\wL_{dp}=\frac{L_{p2}}{(2\pi)^{d\frac{n-k}{2n+d(1-2/p)}}}.$$

Put $d_1(\xi)=|\xi|^k$ and $d_2(\xi)=|\xi|^n$. Then operators $D_1$ and $D_2$ defined above coincide with $(-\Delta)^{k/2}$ and $(-\Delta)^{n/2}$.

\begin{theorem}
Let $2\le p\le\infty$, $k\ge0$, and $n>k$. Then the sharp inequality
\begin{multline*}
\|(-\Delta)^{k/2}x\cd\|_{\ld}\le M_{p2}(k,n,I_0)\|Fx\cd\|_{\lpd}^{\frac{n-k}{n+d(1/2-1/p)}}\\
\times\|(-\Delta)^{n/2}x\cd\|_{\ld}^{\frac{k+d(1/2-1/p)}{n+d(1/2-1/p)}}
\end{multline*}
holds, where $M_{p2}(k,n,I_0)$ is defined by \eqref{mpq} for $2<p\le\infty$ and
$M_{22}(k,n,I_0)=(2\pi)^{(1-k/n)d/2}$.
\end{theorem}

\begin{proof}
The case $p=2$ follows from \cite{MS}. Let $2<p<\infty$. It follows from \eqref{DH} that
$$\|(-\Delta)^{k/2}x\cd\|_{\ld}\le\wL_{dp}\|Fx\cd\|_{\lpd}^{\frac{n-k}{n+d(1/2-1/p)}}
\|(-\Delta)^{n/2}x\cd\|_{\ld}^{\frac{k+d(1/2-1/p)}{n+d(1/2-1/p)}}.$$
We calculate $S_{21}$ and $S_{22}$ which are used in $\wL_{dp}$. We have
\begin{multline*}
S_{21}=\int_{\mathbb R^d}(|t|^{2k}-|t|^{2n})_+^{\frac p{p-2}}\,dt=\int_{|t|<1}|t|^{\frac{2kp}{p-2}}(1-|t|^{2(n-k)})^{\frac p{p-2}}\,dt\\
=\int_0^1\rho^{\frac{2kp}{p-2}}(1-\rho^{2(n-k)})^{\frac p{p-2}}\rho^{d-1}\,d\rho\wII_{12},
\end{multline*}
where $\wII_{12}$ is defined by \eqref{ii}. Changing variables $x=\rho^{2(n-k)}$ we obtain
\begin{multline*}
S_{21}=\frac{\wII_{12}}{2(n-k)}\int_0^1x^{\frac{2kp+d(p-2)}{2(p-2)(n-k)}-1}(1-x)^{\frac p{p-2}}\,dx\\
=\frac{\wII_{12}}{2(n-k)}B\left(\frac{2kp+d(p-2)}{2(p-2)(n-k)},\frac p{p-2}+1\right).
\end{multline*}
The analogous calculations give
\begin{multline*}
S_{22}=\int_{\mathbb R^d}|t|^{2n}(|t|^{2k}-|t|^{2n})_+^{\frac2{p-2}}\,dt\\
=\frac{\wII_{12}}{2(n-k)}B\left(\frac{2kp+d(p-2)}{2(p-2)(n-k)}+1,\frac p{p-2}\right).
\end{multline*}
By properties of beta-function
\begin{align*}
S_{21}&=\frac{\wII_{12}}{2n+d(1-2/p)}B\left(\frac{2kp+d(p-2)}{2(p-2)(n-k)},\frac p{p-2}
\right),\\
S_{22}&=\frac{\wII_{12}(2k+d(1-2/p))}{2(n-k)(2n+d(1-2/p))}B\left(\frac{2kp+d(p-2)}
{2(p-2)(n-k)},\frac p{p-2}\right).
\end{align*}
Hence,
\begin{multline*}
L_{p2}=S_{21}^{-\frac\gamma p}S_{22}^{-\frac{1-\gamma}2}(S_{21}+S_{22})^{1/2}\\
=\gamma^{-\frac\gamma p}(1-\gamma)^{-\frac{1-\gamma}2}\Biggl(
\frac{B\left(\vq\gamma/p+1,\vq(1-\gamma)/2\right)\wII_{12}}{2(n-k)}\Biggr)
^{1/\vq},
\end{multline*}
where $\gamma$ and $\vq$ are defined by \eqref{gq} for $q=r=2$. Consequently,
$$\frac{L_{p2}}{(2\pi)^{d\frac{n-k}{2n+d(1-2/p)}}}=M_{p2}(k,n,I_0).$$

Now let $p=\infty$. Consider the extremal problem
\begin{multline}\label{To}
\|(-\Delta)^{k/2}x\cd\|_{\ld}^2\to\max,\quad\|Fx\cd\|_{L_\infty(\mathbb R^d)}^2\le\delta^2,\\
\|(-\Delta)^{n/2}x\cd\|_{\ld}^2\le1.
\end{multline}
Passing to the Fourier transform we have
\begin{multline}\label{ToFu}
\frac1{(2\pi)^d}\iRd|\xi|^{2k}|Fx(\xi)|^2\,d\xi\to\max,\quad\|Fx\cd\|_{L_\infty(\mathbb R^d)}^2\le\delta^2,\\
\frac1{(2\pi)^d}\iRd|\xi|^{2n}|Fx(\xi)|^2\,d\xi\le1.
\end{multline}
Let $\wx\cd$ be such that
$$F\wx(\xi)=\begin{cases}
\delta,&|\xi|<\sigma,\\
0,&|\xi|\ge\sigma,\end{cases}$$
where $\sigma>0$ is chosen from the condition
$$\frac1{(2\pi)^d}\iRd|\xi|^{2n}|F\wx(\xi)|^2\,d\xi=1.$$
Thus the following equality must be satisfied
$$\delta^2\int_{|\xi|<\sigma}|\xi|^{2n}\,d\xi=(2\pi)^d.$$
Passing to the polar transformation we obtain that $$\sigma=\left(\frac{2^{d-1}\pi^{d/2}\Gamma(d/2)(2n+d)}{\delta^2}\right)^{\frac1{2n+d}}.$$

Consider the Lagrange function for problem \eqref{ToFu}
$$\LL(x\cd,\lambda_1\cd,\lambda_2)=\frac1{(2\pi)^d}\iRd(-|\xi|^{2k}+(2\pi)^d\lambda_1(\xi)
+\lambda_2|\xi|^{2n})|Fx(\xi)|^2\,d\xi.$$
Put $\lambda_2=\sigma^{2(k-n)}$ and
$$\lambda_1(\xi)=\begin{cases}\dfrac1{(2\pi)^d}(|\xi|^{2k}-\lambda_2|\xi|^{2n}),&|\xi|<\sigma,\\
0,&|\xi|\ge\sigma.\end{cases}$$
Then
$$\LL(x\cd,\lambda_1\cd,\lambda_2)=\frac1{(2\pi)^d}\int_{|\xi|\ge\sigma}(-|\xi|^{2k}+
\sigma^{2(k-n)}|\xi|^{2n})|Fx(\xi)|^2\,d\xi\ge0$$
and $\LL(\wx\cd,\lambda_1\cd,\lambda_2)=0$. Consequently, for any admissible function $x\cd$ we have
\begin{multline*}
-\frac1{(2\pi)^d}\iRd|\xi|^{2k}|Fx(\xi)|^2\,d\xi\ge-\frac1{(2\pi)^d}\iRd|\xi|^{2k}
|Fx(\xi)|^2\,d\xi\\
+\iRd\lambda_1(\xi)(|Fx(\xi)|^2-\delta^2)\,d\xi+\lambda_2\biggl(\frac1{(2\pi)^d}\iRd|\xi|^{2n}
|Fx(\xi)|^2\,d\xi-1\biggr)\\
=\LL(x\cd,\lambda_1\cd,\lambda_2)-\delta^2\iRd\lambda_1(\xi)\,d\xi-\lambda_2\\
\ge\LL(\wx\cd,\lambda_1\cd,\lambda_2)-\delta^2\iRd\lambda_1(\xi)\,d\xi-\lambda_2\\
=-\frac1{(2\pi)^d}\iRd|\xi|^{2k}|F\wx(\xi)|^2\,d\xi.
\end{multline*}
It means that the value of problem \eqref{ToFu} (and \eqref{To}) is equal to
$$\frac1{(2\pi)^d}\iRd|\xi|^{2k}|F\wx(\xi)|^2\,d\xi=\delta^2\frac1{(2\pi)^d}\int_{|\xi|<\sigma}
|\xi|^{2k}\,d\xi\\
=M^2\delta^{\frac{4(n-k)}{2n+d}},$$
where
$$M=\frac{(2n+d) ^{\frac{k+d/2}{2n+d}}}{\sqrt{2k+d}}\left(\frac1{2^{d-1}\pi^{d/2}\Gamma(d/2)}\right)^{\frac{n-k}
{2n+d}}.$$
Thus,
$$\|(-\Delta)^{k/2}x\cd\|_{\ld)}\le M\delta^{\frac{2(n-k)}{2n+d}}.$$
It follows from this inequality (as in the proof of Theorem~\ref{TT1}) that
$$\|(-\Delta)^{k/2}x\cd\|_{\ld}\le M\|Fx\cd\|_{L_\infty(\mathbb R^d)}^{\frac{n-k}{n+d/2}}
\|(-\Delta)^{n/2}x\cd\|_{\ld}^{\frac{k+d/2}{n+d/2}}.$$
It remains to note that $M=M_{\infty2}(k,n,I_0)$.
\end{proof}

Consider the case when $D_1=D^\alpha$, $D_2=(-\Delta)^{n/2}$ and $p=\infty$.

\begin{theorem}
Let $n>|\alpha|$. Then the sharp inequality
\begin{equation}\label{Las}
\|D^\alpha x\cd\|_{\ld}\le\wM_d\|Fx\cd\|_{L_\infty(\mathbb R^d)}^{\frac{n-|\alpha|}{n+d/2}}
\|(-\Delta)^{n/2}x\cd\|_{\ld}^{\frac{|\alpha|+d/2}{n+d/2}}
\end{equation}
holds, where
$$\wM_d=\frac{(n+d/2) ^{\frac{|\alpha|+d/2}{2n+d}}}{\sqrt{|\alpha|+d/2}}J_1^{\frac{n-|\alpha|}
{2n+d}}$$
and
$$J_1=\frac{\Gamma\left(\alpha_1\dfrac{n+d/2}{n-|\alpha|}+1/2\right)\ldots
\Gamma\left(\alpha_d\dfrac{n+d/2}{n-|\alpha|}+1/2\right)}{(2\pi)^d\Gamma\left(n\dfrac{|\alpha|+d/2}{n-|\alpha|}
\right)}.$$
\end{theorem}

\begin{proof}
Consider the extremal problem
\begin{multline}\label{To1}
\|D^\alpha x\cd\|_{\ld}^2\to\max,\quad\|Fx\cd\|_{L_\infty(\mathbb R^d)}^2\le\delta^2,\\
\|(-\Delta)^{n/2}x\cd\|_{\ld}^2\le1.
\end{multline}
Passing to the Fourier transform we have
\begin{multline}\label{ToFu1}
\frac1{(2\pi)^d}\iRd|\xi^{2\alpha}||Fx(\xi)|^2\,d\xi\to\max,\quad\|Fx\cd\|_{L_\infty(\mathbb R^d)}^2\le\delta^2,\\
\frac1{(2\pi)^d}\iRd|\xi|^{2n}|Fx(\xi)|^2\,d\xi\le1.
\end{multline}
Let $\wx\cd$ be such that
$$F\wx(\xi)=\begin{cases}
\delta,&|\xi^{2\alpha}|>\lambda|\xi|^{2n},\\
0,&|\xi^{2\alpha}|\le\lambda|\xi|^{2n},\end{cases}$$
where $\lambda>0$ is chosen from the condition
$$\frac1{(2\pi)^d}\iRd|\xi|^{2n}|F\wx(\xi)|^2\,d\xi=1.$$
Thus the following equality is valid
$$\delta^2\int_{|\xi^{2\alpha}|>\lambda|\xi|^{2n}}|\xi|^{2n}\,d\xi=(2\pi)^d.$$
Passing to the polar transformation we obtain
$$\delta^2\int_{\Pi_{d-1}}J(\omega)\,d\omega\int_0^{\Phi_2(\omega)}
\rho^{2n+d-1}\,d\rho=(2\pi)^d,$$
where
$$\Phi_2(\omega)=\lambda^{-\frac1{2(n-|\alpha|)}}\Phi\left(\omega,\frac{\alpha_1}{n-|\alpha|}+1,
\ldots,\frac{\alpha_d}{n-|\alpha|}+1\right).$$
Hence,
$$\lambda=\left(\frac{\delta^2J_1}{n+d/2}\right)^{\frac{n-|\alpha|}{n+d/2}}.$$

Consider the Lagrange function for problem \eqref{ToFu1}
$$\LL(x\cd,\lambda_1\cd,\lambda_2)=\frac1{(2\pi)^d}\iRd(-|\xi^{2\alpha}|+(2\pi)^d\lambda_1(\xi)
+\lambda_2|\xi|^{2n})|Fx(\xi)|^2\,d\xi.$$
Put $\lambda_2=\lambda$ and
$$\lambda_1(\xi)=\begin{cases}\dfrac1{(2\pi)^d}(|\xi^{2\alpha}|-\lambda_2|\xi|^{2n}),&|\xi^{2\alpha}|>
\lambda|\xi|^{2n},\\
0,&|\xi^{2\alpha}|\le\lambda|\xi|^{2n}.\end{cases}$$
Then
$$\LL(x\cd,\lambda_1\cd,\lambda_2)=\frac1{(2\pi)^d}\int_{|\xi^{2\alpha}|\le\lambda|\xi|^{2n}}
(-|\xi^{2\alpha}|+\lambda|\xi|^{2n})|Fx(\xi)|^2\,d\xi\ge0$$
and $\LL(\wx\cd,\lambda_1\cd,\lambda_2)=0$. Consequently, for any admissible function $x\cd$ we have
\begin{multline*}
-\frac1{(2\pi)^d}\iRd|\xi^{2\alpha}||Fx(\xi)|^2\,d\xi\ge-\frac1{(2\pi)^d}\iRd|\xi^{2\alpha}|
|Fx(\xi)|^2\,d\xi\\
+\iRd\lambda_1(\xi)(|Fx(\xi)|^2-\delta^2)\,d\xi+\lambda_2\biggl(\frac1{(2\pi)^d}\iRd|\xi|^{2n}
|Fx(\xi)|^2\,d\xi-1\biggr)\\
=\LL(x\cd,\lambda_1\cd,\lambda_2)-\delta^2\iRd\lambda_1(\xi)\,d\xi-\lambda_2\\
\ge\LL(\wx\cd,\lambda_1\cd,\lambda_2)-\delta^2\iRd\lambda_1(\xi)\,d\xi-\lambda_2\\
=-\frac1{(2\pi)^d}\iRd|\xi^{2\alpha}||F\wx(\xi)|^2\,d\xi.
\end{multline*}
Thus, the value of problem \eqref{ToFu1} (and \eqref{To1}) is equal to
$$\frac1{(2\pi)^d}\iRd|\xi^{2\alpha}||F\wx(\xi)|^2\,d\xi=\delta^2\frac1{(2\pi)^d}
\int_{|\xi^{2\alpha}|>\lambda|\xi|^{2n}}
|\xi^{2\alpha}|\,d\xi\\
=\wM_d^2\delta^{\frac{4(n-|\alpha|)}{2n+d}}.$$
Hence, for any admissible function $x\cd$
$$\|D^\alpha x\cd\|_{\ld}\le\wM_d\delta^{\frac{n-|\alpha|}{n+d/2}}.$$
This implies \eqref{Las}.
\end{proof}








\end{document}
